Author’s response to reviews

Title: CHERISH (Collaboration for Hospitalised Elders Reducing the Impact of Stays in Hospital): Protocol for a multi-site improvement program to reduce geriatric syndromes in older inpatients

Authors:

Alison Mudge (Alison.Mudge@health.qld.gov.au) (amudge@westnet.com.au)
Merrilyn Banks (Merrilyn.Banks@health.qld.gov.au)
Adrian Barnett (a.barnett@qut.edu.au)
Irene Blackberry (i.blackberry@latrobe.edu.au)
Nicholas Graves (n.graves@qut.edu.au)
Theresa Green (theresa.green@qut.edu.au)
Gillian Harvey (gill.harvey@adelaide.edu.au)
Ruth Hubbard (rhubbard1@uq.edu.au)
Sharon Inouye (AgingBrain@hsl.harvard.edu)
Sue Kurrle (susan.kurrle@sydney.edu.au)
Kwang Lim (kwang.lim@mh.org.au)
Prue McRae (Prue.McRae@health.qld.gov.au)
Nancye Peel (n.peel@uq.edu.au)
Jessica Suna (Jessica.Suna@health.qld.gov.au)
Adrienne Young (adrienne.young@health.qld.gov.au)

Version: 1 Date: 10 Dec 2016

Author’s response to reviews:

Thankyou, I have changed the background header and removed the tables and figures as requested